Author's response to reviews

Title: Pakistanis living in Oslo have lower serum 1,25-dihydroxyvitamin D levels but higher serum ionized calcium levels compared with ethnic Norwegians. The Oslo Health Study

Authors:

- Kristin Holvik (kristin.holvik@medisin.uio.no)
- Haakon E Meyer (h.e.meyer@medisin.uio.no)
- Anne Johanne Sogaard (anne.johanne.sogaard@fhi.no)
- Egil Haug (egil.haug@medisin.uio.no)
- Jan A Falch (j.a.falch@medisin.uio.no)

Version: 3 Date: 17 October 2007

Author's response to reviews:

Please find attached our latest version of the manuscript with formatting changes as requested by your journal. We have made the following changes:
- the full stop has been removed from the end of the title. This has also been updated in the submission system.
- the Ethics information has been moved to the end of the Methods section.
- all text is black